Author's response to reviews

Title: Hip and fragility fracture prediction by 4-item clinical risk score and mobile heel BMD: a women cohort study

Authors:

Daniel M Albertsson (daniel.albertsson@ltkronoberg.se)
Dan Mellström (dan.mellstrom@vgregion.se)
Christer Petersson (christer.petersson@ltkronoberg.se)
Hans Thulesius (hans.thulesius@ltkronoberg.se)
Robert Eggertsen (robert.eggertsen@vgregion.se)

Version: 4 Date: 18 November 2009

Author's response to reviews:

Editorial Coordinator
Editorial office
BMC Musculoskeletal Disorders

Thank you for your response on our manuscript “Hip and fragility fracture prediction by 4-item clinical risk score and mobile heel BMD: a women cohort study”. We now return our revised manuscript, for further evaluation and possible publication in the BMC Musculoskeletal Disorders.

Send you now our comments in our gramatically updated coverletter:

You did request further information about “informed consent” in our manuscript. Each participant received oral and written study information and approved study participation by returning the questionnaire and undergoing BMD assessment. The study design was approved by the Regional Ethics Committee at Lund University, evaluating the severity of the experimental procedure against the value of knowledge gained as well. Also, the local radiation protection committee at Växjö Central Hospital approved the DXL and DXA screening investigation. Therefore, on page 9-10 we changed the text below heading “Ethics” into:

“The Regional Ethics Committee at Lund University approved the study (LU 406-00). Also, the local radiation protection committee at Växjö Central Hospital approved the DXL and DXA screening. Each participant received oral and written study information and approved study participation by returning the questionnaire and undergoing BMD assessment.”

Also, on page 4: we changed the heading “Introduction” to “Background”.

Yours sincerely

Daniel Albertsson, Dan Mellström, Christer Petersson, Hans Thulesius and Robert Eggertsen